Author's response to reviews

Title: Ferritin levels and risk of metabolic syndrome: meta-analysis of observational studies

Authors:

Victoria Abril-Ulloa (victoria.abril@ucuenca.edu.ec)
Gemma Flores-Mateo (gflores@idiapigol.org)
Rosa Solà-Alberich (rosa.sola@urv.cat)
Begona Manuel-y-Keenoy (begona.manuelykeenoy@uantwerpen.be)
Victoria Arija (victoria.arija@urv.cat)

Version: 2  Date: 22 March 2014

Author's response to reviews:

21st March, 2014

The Editorial Office
BMC Public Health

Dear Editor,

Please find, attached, our manuscript entitled “Ferritin levels and risk of metabolic syndrome: meta-analysis of observational studies (MS: 1277118093116070)” with the suggested modifications, to be considered in BMC Public Health (Section: Disease epidemiology-chronic).

Below you'll find the answers to your comments and the indications of the changes we have made, which are highlighted in green in the text and in the tables of the paper.

Yours sincerely,

Victoria Arija, Prof